Author’s response to reviews

Title: Targeted next generation sequencing with an extended gene panel does not impact variant detection in mitochondrial diseases

Authors:

Morgane Plutino (plutino.m@chu-nice.fr)
Annabelle Chaussenot (chaussenot.a@chu-nice.fr)
Cécile Rouzier (rouzier.c@chu-nice.fr)
Samira Ait-El-Mkadem (saadi.s@chu-nice.fr)
Konstantina Fragaki (fragaki.k@chu-nice.fr)
Véronique Paquis-Flucklinger (paquis@unice.fr)
Sylvie Bannwarth (bannwarth.s@chu-nice.fr)

Version: 1 Date: 08 Nov 2017

Author’s response to reviews:

Dear Editor,

Please accept kindly the reviewed manuscript and tables, with all the modifications made in response to yours comments.

This is an article of clinical research and our consents are provided by the french alliance against rare disease "fondation maladies rares". Those consents forms are used in diagnostic and in clinical research. French legislation does not imposed ethical comittee approuvement in this kind of retrospective study.

Best regards